% $Id: solution.tex 4241 2006-11-15 14:43:42Z gokhale $


\section{Declarative Network QoS Provisioning Capabilities in Component Middleware}
\label{sec:solution}

This section describes the novel enhancements we made to our QoS
provisioning framework for component-based DRE systems.  First, we
provide an overview of our existing QoS provisioning framework.  Next
we describe our enhancements to this framework to enable network-level
QoS provisioning.

% overview
\input{overview}

% MDE
\input{netqos_mde}

% planner
\input{netqos_planner}

% policy framework
\input{policy}


%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "icdcs07"
%%% End: 
